CS150 Newton 7.1. 1

Outline

m Last time:

Ü Adders and subtracters : Half adder, full adder

Ü Adder organizations: ripple,carry look-ahead(CLA), carry

bypass

Ü Full adder as a full subtracter

Ü Sign-magnitude

Ü One’s & Two’s Complement

m This lecture:

Ü Floating Point Formats

Ü IEEE Formats

Ü Multiplication

Ü Division

Ü The Arithmetic Logic Unit (ALU)

Ü Bandwidth &Latency

Ü Scheduling & Allocation,Pipelining

Parallel Addition with Accumulator

n-bit

parallel

adder n-bit

parallel

register

B

clock

Out

Q

D

B

A S

Cout

Overflow

"M+" operation

Fractions

D = S d i .r^i

i=-n p-1

The value of a number in any radix is given by:

m Where di represents the ith digit of the number,